Author's response to reviews

Title: Treatment outcomes in deep sternal wound infection after cardiac surgery; Vacuum-assisted closure versus conventional treatment

Authors:

Hayati Deniz (hayatideniz@gmail.com)
Gokhan Gokaslan (cerrah06@yahoo.com)
Yavuz Arslanoglu (dr.arslanoglu@gmail.com)
Ozerdem Ozcaliskan (ozerdemo@yahoo.com)
Gokalp Guzel (drgokalpguzel@hotmail.com)
Alptekin Yasim (alpyasim@yahoo.com)
Hasim Ustunsoy (hustunsoy@yahoo.com)

Version: 6 Date: 10 June 2012

Author's response to reviews:

# have revised my öanuscript due to the rewievers comments thank you